Author's response to reviews

Title: When Female Circumcision Comes to the West: Attitudes Toward the Practice among Somali Immigrants in Oslo

Authors:

Abdi A Gele Gele, AA (suuleycg@hotmail.com)
Elise B Johansen Johansen EB (JohansenE@who.int)
Mona I Huage Hauge MI (m.i.hauge@nkyts.unirand.no)
Johanne Sundby Sundby J (johanne.sundby@medisin.uio.no)

Version: 4 Date: 19 August 2012

Author's response to reviews:

Editor's comments

1. In the paragraph on limitations the fact that self-reported type of FGM may not be reliable information and that a validation study is under way (this is included in the response to reviewer one but not included in the text)

We added the statement suggested by the editor into the limitation section: look page 11, paragraph 3, line 3-6

2. Revise the format of the reference list

References were revised thoroughly

3. Add some English copyediting (for example Haemorrhage in table 2)

The manuscript was copy edited and proof-read by native-english speaker